Author's response to reviews

Title: Appendicectomies in Albanians in Greece: outcomes in a highly mobile immigrant patient population

Authors:
- Athina Tatsioni (athina73@otenet.gr)
- Antonia Charchanti (me00527@cc.uoi.gr)
- Evangelia Kitsiou (jioannid@cc.uoi.gr)
- John P.A. Ioannidis (jioannid@cc.uoi.gr)

Version: 4 Date: 26 Jun 2001

Dear Editors:

thank you again for the excellent peer-review of our manuscript and for its final acceptance. We have added the requested competing interests section. The title is unchanged - given the peculiarity of the study design, using the "case control" nomenclature might be misleading to many readers.

Sincerely,

John P.A. Ioannidis, MD